\input{ThisWeek.tex}
\newcommand{\duedate}{Thursday, September 11}
\input{../TemplateWeek/CheckHomeWork.tex}
\input{Exercises.tex}

%\booltrue{teacher}

\begin{enumerate}
\item See the next one.
\item This is $\frac{2}{9}$ times the answer to the first one.
\item This is $10$ divided by the answer to the first one
\item You can check on a calculator: $\frac{A(1.01)-A(1)}{A(1)}*100$ is close to an integer, and it should be the integer you got as your answer.
\item Plug into a calculator.  The difference between the correct answer and your estimate should be about $0.0000015$.
\end{enumerate}

\end{document}